

Citation: Seifert SN, Fischer RJ, Kuisma E, Badzi Nkoua C, Bounga G, Akongo M-J, et al. (2022) *Zaire ebolavirus* surveillance near the Bikoro region of the Democratic Republic of the Congo during the 2018 outbreak reveals presence of seropositive bats. PLoS Negl Trop Dis 16(6): e0010504. https:// doi.org/10.1371/journal.pntd.0010504

Editor: Waleed Saleh Al-Salem, Saudi Ministry of Health, SAUDI ARABIA

Received: June 17, 2021

Accepted: May 16, 2022

Published: June 22, 2022

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: Relevant data are in the manuscript or available with the associated GenBank accession numbers (MZ265505-MZ265622) provided in the main text. Data from the anti-ZEBOV glycoprotein ELISA is located in S2 Table.

Funding: This work was supported by the Intramural Research Program of the National Institute of Allergy and Infectious Diseases, National Institutes of Health (Grant R15AI089419, RESEARCH ARTICLE

Zaire ebolavirus surveillance near the Bikoro region of the Democratic Republic of the Congo during the 2018 outbreak reveals presence of seropositive bats

Stephanie N. Seifert^{1,2*}, Robert J. Fischer², Eeva Kuisma³, Cynthia Badzi Nkoua⁴, Gerard Bounga³, Marc-Joël Akongo³, Jonathan E. Schulz², Beatriz Escudero-Pérez^{5,6}, Beal-Junior Akoundzie⁴, Vishnou Reize Bani Ampiri⁴, Ankara Dieudonne⁷, Ghislain Dzeret Indolo⁴, Serge D. Kaba³, Igor Louzolo⁴, Lucette Nathalie Macosso⁴, Yanne Mavoungou⁴, Valchy Bel-bebi Miegakanda⁴, Rock Aimé Nina^{7†}, Kevin Tolovou Samabide^{4,8}, Alain I. Ondzie³, Francine Ntoumi^{8,9,10}, César Muñoz-Fontela⁵, Jean-Vivien Mombouli⁴, Sarah H. Olson¹¹, Chris Walzer^{11,12}, Fabien Roch Niama⁴, Vincent J. Munster²

 Paul G. Allen School for Global Health, College of Veterinary Medicine, Washington State University, Pullman, Washington, United States, 2 Virus Ecology Section, Laboratory of Virology, Rocky Mountain Laboratories, National Institutes of Allergy and Infectious Diseases, National Institutes of Health, Hamilton, Montana, United States, 3 Wildlife Health Program, Wildlife Conservation Society, Brazzaville, Republic of the Congo, 4 Département de la Recherche et de la Production, Laboratorie National de Santé Publique, Brazzaville, Republic of the Congo, 5 Bernhard Nocht Institute for Tropical Medicine, Hamburg, Germany, 6 German Centre for Infection Research (DZIF), Partner Site Hamburg-Luebeck-Borstel, Germany,
7 Direction de la Santé Animale, Ministére de L'Agriculture et de L'Élevage, Brazzaville, Republic of the Congo, 8 Faculty of Sciences and Techniques, Université Marien N'Gouabi, Brazzaville, Republic of the Congo, 9 Fondation Congolaise pour la Recherche Médicale, Brazzaville, Republic of the Congo, 10 Institute of Tropical Medicine, University of Tübingen, Tübingen, Germany, 11 Health Program, Wildlife Conservation Society, New York, New York, United States, 12 Research Institute of Wildlife Ecology, University of Veterinary Medicine, Vienna, Austria

† Deceased.

* stephanie.seifert@wsu.edu

Abstract

On the 8th of May, 2018, an outbreak of Ebola virus disease (EVD) was declared, originating in the Bikoro region of the Democratic Republic of the Congo (DRC) near the border with neighboring Republic of the Congo (ROC). Frequent trade and migration occur between DRC and ROC-based communities residing along the Congo River. In June 2018, a field team was deployed to determine whether *Zaire ebolavirus* (Ebola virus (EBOV)) was contemporaneously circulating in local bats at the human-animal interface in ROC near the Bikoro EVD outbreak. Samples were collected from bats in the Cuvette and Likouala departments, ROC, bordering the Équateur Province in DRC where the Bikoro EVD outbreak was first detected. EBOV genomic material was not detected in bat-derived samples by targeted quantitative reverse transcription-polymerase chain reaction or by family-level consensus polymerase chain reaction; however, serological data suggests recent exposure to EBOV in bats in the region. We collected serum from 144 bats in the Cuvette department with 6.9% seropositivity against the EBOV glycoprotein and 14.3% seropositivity for serum collected from 27 fruit bats and one Molossinae in the Likouala department. We conclude that

VJM) and the Ministry of Health and Population, Republic of the Congo-Brazzaville (JVM and FRN). The work was partly supported by PANDORA and EPIRISK projects (EDCTP Grant Agreement RIA2016E-1609 and RIA2018F – 2082- EPIRISK-EBOV, FN and CMF), the United States Fish and Wildlife Service and the Arcus Foundation (SHO and CW). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

proactive investment in longitudinal sampling for filoviruses at the human-animal interface, coupled with ecological investigations are needed to identify EBOV wildlife reservoirs.

Author summary

Zaire ebolavirus (Ebola virus, (EBOV)), first characterized more than 40 years ago, is a causative agent of Ebola virus disease (EVD) which is associated with viral hemorrhagic fever. EBOV is thought to be transmitted to humans through contact with wildlife. Though bats are suspected to be the hosts that maintain EBOV in nature, it is currently not known which bat species serve as the reservoir. Identifying the wildlife species that maintain EBOV in nature would facilitate the development of strategies to mitigate the risk of EBOV transmission to humans from wildlife. In this manuscript, we describe the results of surveillance efforts to detect Ebola virus in wildlife near a contemporaneous outbreak of EBOV disease in humans. We detected evidence of recent EBOV infection in bats near the location of the EVD outbreak through serological techniques, but we did not find evidence of active EBOV infection in bats. These data support the need for more proactive surveillance in wildlife in EBOV endemic regions.

Introduction

More than forty years have passed since the characterization of *Zaire ebolavirus* (Ebola virus (EBOV)) as a causative agent of Ebola virus disease (EVD), and still the wildlife reservoir remains elusive. Since that time there have been dozens of EVD outbreaks caused by multiple viruses in the *Ebolavirus genus* including Ebola virus, Sudan virus, and Bundibugyo virus, with EVD outbreaks primarily originating in Central Africa. Bats have been implicated as the most likely wildlife reservoir of EBOV [1] and other related filoviruses [2,3,4,5,6,7]. Despite many efforts to identify the reservoir of EBOV, only a single group has detected and published on EBOV nucleic acids in wild bats [1]. Following extensive sampling in a region where several EBOV spillover events had been detected in humans and in great apes from 2001-2003 at the border between northern Gabon and Republic of Congo (ROC), Leroy et al. (2005) detected Ebola virus nucleic acids in 13 bats belonging to three species of fruit bat, Hypsignathus monstrosus, Epomops franqueti, and Myonycteris torquata [1]. Since that time there have been no new EVD outbreaks in Gabon or ROC and no published data has replicated the findings of Leroy et al. 2005[1]. Identifying the reservoir or reservoirs of EBOV in the Congo basin continues to elude researchers, hindering the effort to identify the ecological and socio-economic factors that contribute to spillover events.

Outbreaks of EVD originating in Équateur Province, Democratic Republic of the Congo (DRC) have been reported three times in recent years with an EVD outbreak Jul—Nov, 2014, resulting in 66 EVD cases and 49 deaths [8], May—Jul, 2018, in Bikoro, DRC, resulting in 54 EVD cases and 33 deaths [9,10], and May–Nov, 2020 with 130 cases and 55 deaths reported [11]. The Équateur Province in northwestern DRC is adjacent to the Congo River which delineates the border between DRC and neighboring Republic of the Congo (ROC); there is frequent trade and migration between the countries along the Congo River.

Following the declaration of an outbreak of EVD on the 8th of May, 2018, in the market town Bikoro, Équateur Province, DRC, a field team was deployed to the municipalities, Mossaka and Liranga, ROC, near the Bikoro EVD outbreak in DRC to conduct surveillance in bats

for EBOV (Fig 1). We prioritized sampling from synanthropic bat species including bats in the subfamily Molossinae, *Mops condylurus* (Angolan free-tailed bat) and *Chaerephon pumilus* (little free-tailed bat), which are commonly associated with human dwellings in central Africa. *M. condylurus* bats have been implicated as a plausible source of the 2013 EBOV spillover in Guinea [12], while genomic material for a related Ebolavirus, Bombali virus, has been recovered from both *M. condylurus* and *C. pumilus* bats [4, 13, 14].

Methods

Ethics statement

All field work was performed under the approval of L'Institut National de Recherche Forestiére, Brazzaville, ROC (Permit No. 260) and under approval of the Rocky Mountain Laboratories, Division of Intramural Research, National Institute of Allergy and Infectious Diseases, National Institutes of Health, Institutional Animal Care and Use Committee (Animal Study Protocol #2018-15-F) and Institutional Biosafety Committee (IBC).

Sampling

The field team arrived in Mossaka, ROC, and began sampling bats on the 20th of June, 2018, with researchers from the National Public Health Laboratory, Congo-Brazzaville, the Wildlife Health Program, Wildlife Conservation Society, Congo-Brazzaville, and the National Institutes of Health, United States.

Mist nets were deployed in Mossaka, ROC, near the local hospital (Lat: -1.224809, Long: 16.797453) where the Likouala tributary branches from the Congo River known for fish production. Mossaka was selected as a sampling location due to the proximity to the Bikoro EVD outbreak, observed abundance of bats roosting in human structures, large population relative to surrounding municipalities, and the importance of the town as a port for trade on the floating marketplace between ROC and DRC. The town is structured with residential blocks bordering the river with narrow, sandy paths frequented by pedestrians and motorcyclists. We noted the presence of several free-roaming cats, dogs, chickens, goats, and two pigs. Bats were actively hunting mosquitoes and other insects aggregating near artificial light sources throughout the town of Mossaka. In order to minimize both the biohazard risk and potential disruption to the town, we set the mist nets near sources of artificial light in the central courtyard of the Mossaka Hospital after the gates closed at dusk. As the Mossaka Hospital was the largest source of artificial light in the town after 22:00, insectivorous bat hunting activity was high in this clearing. Mist nets were set until 10 bats were captured, then closed until 8 of those bats were sampled and released. We aimed to process approximately 25 bats per sampling night at this location. Mist nets were also deployed in the fishing village, Liranga, ROC, (Lat: -0.655914, Long: 17.606579) which is adjacent to Bikoro, DRC, the location of the EVD outbreak (Fig 1). Liranga was selected as a sampling location as it is geographically situated near the epicenter of the Bikoro EVD outbreak and the open clearings of Liranga are surrounded by disturbed forest which provided an opportunity to sample from fruit bats. The mist nets were set up in a clearing near the riverbank. There were few free-roaming livestock and companion animals in the village, including pigs, goats, chickens, dogs, and cats, as well as an outdoor market with primarily fish. We did not see wild game at the time of sampling in the village. In Liranga, the mist nets were set up at 19:00 and checked every 20 minutes for bats for two nights with 15 bats captured one night and 16 bats captured the second night.

Oral/nasal, urogenital, and rectal swabs were collected into 1mL of viral transport media for each bat sampled, swab samples were then stored the same night in portable -80°C freezers or liquid nitrogen dry shippers before long-term storage at -45°C. Swabs were not collected from



Fig 1. Locations of historical Ebola virus disease outbreaks in Africa associated with Zaire ebolavirus (Ebola virus, red), Sudan virus (yellow), or Bundibugyo virus (blue). Inset map shows the sampling locations in Mossaka and Liranga, Republic of Congo, relative to the 2018 Ebola virus disease outbreak in Bikoro, Democratic Republic of Congo. A total of 31 bats were sampled at the Liranga field sites and 165 bats were sampled at the Mossaka field site. Base layer map data sourced through Natural Earth http://www.naturalearthdata.com/.

https://doi.org/10.1371/journal.pntd.0010504.g001

11 bats which were lethally sampled in Liranga from which tissue samples were collected. Blood was collected from 144 bats at the Mossaka field site and 28 bats at the Liranga field site. Blood or blood in pH neutral phosphate buffered saline (PBS), for smaller bats, was aliquoted for molecular surveillance and frozen in -80°C freezers or liquid nitrogen dry shippers the same night before long-term storage at -45°C. Nonlethal blood volume collected from Molossinae bats varied from 5–50 uL depending on size of bat (juvenile Molossinae bats weighed as little as 7 g) and the blood flow which can be influenced by several factors including hydration and temperature. Blood volume collected for each individual was recorded for and transferred from the capillary tube into 500 uL of PBS with 140 uL immediately transferred to Qiagen Buffer AVL (Qiagen, Germantown, MD USA) for ribonucleic acid (RNA) extraction as per our institutional biosafety committee-approved protocol for working with field-collected samples which may contain a select agent. The remaining blood or blood in PBS was stored in serum separator tubes overnight before centrifugation at 1000 X g for 10 minutes. Serum was then transferred by pipette into a sterile microcentrifuge tube and frozen -80°C freezers or liquid nitrogen dry shippers the same night before long-term storage at -45°C.

Molecular surveillance

Sample inactivations were performed with Qiagen Buffer AVL (Qiagen, Germantown, MD USA) and Ethanol as described in Haddock et al. 2016 [15] in a Class III biosafety cabinet located in the Laboratoire National de Santé Publique, Brazzaville, ROC. Oral/nasal, urogenital, and rectal swab samples were pooled for each individual bat sampled, 50 uL each with 140 uL then transferred for the RNA extraction, and extracted using the QIAamp Viral RNA extraction kit (Qiagen, Germantown, MD USA). Whole blood was extracted using the QIAamp Viral RNA extraction kit with an additional wash step (Qiagen, Germantown, MD USA). RNA was extracted from the liver, lung, and spleen using the RNeasy extraction kit (Qiagen, Germantown, MD USA) from 23 bats that were lethally sampled including 9 Chaerephon pumilus and 4 Mops condylurus bats from the Mossaka field site, and 5 Epomops franqueti, 2 Hypsignathus monstrosus, and 3 Micropteropus pusillus bats from the Liranga field site. A multiplexed, quantitative real-time polymerase chain reaction (qRT-PCR) was used to detect presence of EBOV and Bombali virus nucleic acids, modified from de Wit et al. 2016[16] to include primers and probes (Integrated DNA Technologies, Inc. Coralville, Iowa, USA) targeting the L gene for both Ebola virus (5'-CAGCCAGCAATTTCTTCCAT-3, 5'-TTTCGGTTGCTGTTTCTGTG-3', FAM-ATCATTGGC/ZEN/RTACTGGAGGAG-CAG-BHQ, FAM-TCATTGGCG/ZEN/TACTGGAGGAGCAGG-BHQ), Bombali virus (5'-TCTCGACGAAGGTCATTAGCGA-3', 5'-TTGCTCTGGTACTCGCTTGGT-3', FAM-TGCTGGGATGCTGTCTTTGAGCCT-BHQ) and bat HPRT (5'-AGATGGT-CAAGGTCGCAAG-3', 5'-CCTGAAGTATTCATTATAGTCAAGGG-3', HEX-ACTTTGGTTGGATTTGAAATTCCAGACAAGTTTG-BHQ) as an internal control. Briefly, 5 µL of extracted RNA was added per reaction with 20 µL of LightCycler 480 RNA Master Hydrolysis Probes reaction kit master mix (Roche, Indianapolis, IN; Product #04991885001). Thermal cycling included an initial reverse transcription step at 63°C for 3 minutes, an initial denaturation step at 95°C for 30 seconds, followed by 40 cycles of 95°C for 10 seconds, 60°C for 30 seconds, and 72°C for 6 seconds completed on a Magnetic Induction qPCR Cycler (MIC, Bio Molecular Systems, El Cajon, CA USA). A filovirus family-level reverse transcription-polymerase chain reaction (RT-PCR) was used to detect presence of filovirus nucleic acids as described in Schulz et al. 2020[17]. Full or partial cytochrome b was sequenced from 117 of 196 bats (accession numbers MZ265505-MZ265622) sampled to confirm morphological identification from the field. Species were confirmed based on greater than 98% sequence identity across 450 bp or 1,000 bp length sequences depending on whether longer amplicons could be recovered from the sample. Full or partial *cytochrome b* sequence was recovered for each bat using 3 µL of cDNA template, primers (Ducroz et al. 2001[18], long amplicons: 5'-ACCAATGACATGAAAAATCATCGTT-3', 5'-TCTCCATTTCTGGTTTTACAAGAC-3'; or short amplicons targeting Molossinae bats: 5'-TACCATGAGGACAAATATC-3', 5'-CAAC-TAGCAGTCAAAATAGA-3', 5'-TAGATAGGACTAGGGCTAGT-3') at 0.2 µM each in a 25 µL reaction with the CloneAmp HiFi PCR premix (Takara Bio USA, Inc, Mountain View, CA) and amplified with 35 cycles of 3-step PCR at 98°C for 10 seconds, 52°C for 5 seconds, and 72°C for 5 seconds.

Serological surveillance

A commercial human anti-ZEBOV glycoprotein (GP) enzyme linked immunosorbent assay (ELISA) (Alpha Diagnostics, San Antonio, TX USA) was modified as follows; 100 μ L bat serum diluted to 1:100 in PBS was added in duplicate to each well of the human anti-ZEBOV ELISA GP Immunoglobulin G (IgG) kit (#AE-320620-5) and incubated at room temperature for 1 hour, rinsed four times with the kit wash buffer before adding horseradish peroxidase-conjugated bethyl goat anti-bat IgG antibody at a 1:50,000 dilution and incubating for 30 minutes in an air-conditioned room at room temperature. The antibody was then washed four times with the kit wash buffer 100 μ L of stop solution was added and allowed to incubate at room temperature for 15 min before 100 μ L of stop solution was added and the optical density was measured at 450 nm on a BioRad iMark microplate reader. Threshold for seropositivity was determined as 3 times the standard deviation plus the mean OD of naive serum from captive bred *Rousettus aegyptiacus* bats.

Results

A total of 196 bats were sampled; 165 bats from the Molosinnae subfamily sampled in Mossaka, ROC, and 30 fruit bats and a single bat from the Molosinnae subfamily was sampled in Liranga, ROC. No EBOV or filovirus genomic material was detected in any samples by targeted qRT-PCR or by pan-filovirus RT-PCR including lung, liver, and spleen samples from 23 lethally sampled bats, 172 blood samples, and pooled oral/nasal, urogenital, and rectal swabs from 185 individuals (S1 Table). We recovered serum samples from 172 of 196 bats. Serum samples were screened for reactivity against the EBOV glycoprotein; 8.1% of all bats sampled were seropositive with a cut-off of the mean OD plus 3 times the standard deviation of the naive controls (Table 1, S2 Table). We found higher seroprevalence among fruit bats in Liranga relative to the insectivorous bats sampled in Mossaka; however, fewer bats were sampled in Liranga (N = 28 and N = 144, respectively, Table 1). None of the *Micropteropus pusillus* bats were seropositive (N = 12), while both juvenile *Hypsignathus monstrosus* bats were seropositive and two of twelve adult *Epomops franqueti* bats were seropositive (Table 1). As we have fewer than 15 individuals sampled per species at the Liranga site, we did not perform bivariate analyses on this dataset. Cytochrome b sequence was not recovered for 49 of 145 Molossinae bats sampled, hindering species-level delineation for that group. Among the 145 Molossinae (Mops and Chaerephon) bats sampled in total, 54 were juvenile bats. While a lower proportion of juvenile Molossinae bats were seropositive relative to the adults (1 of 54 juvenile bats and 9 of 91 adult bats sampled, Table 1) the difference was not statistically significant by Chi-square Test and Fisher's test executed in Prism version 9.1.1.

Discussion

More than forty years have passed since the characterization of EBOV, including decades of ecological surveillance efforts, and still the wildlife reservoir remains elusive. Several surveillance efforts have reported on Ebolavirus seroprevalence in bats in Africa [19,20,21,22,23]. Using a multiplexed Luminex assay, LaCroix *et al.* (2021) reported a seroprevalence of 0.3–2.8% against at least one Ebolavirus antigen in the Bikoro region for bats sampled contemporaneously with this study, and none of the insectivorous bats were seropositive [23]. The two studies are complementary in sampling efforts with La Croix *et al.* (2021) having broadly sampled from frugivorous bats in the Équateur Province (>95% of samples) during the Bikoro EVD outbreak [23], while we sampled extensively from synanthropic insectivorous bats, *M. condylurus* and *C. pumilus*, which were found roosting in the wall cavities of human dwellings throughout Mossaka (S1 Fig, S1 Video). *M. chaerephon* and *C. pumilus* are widely distributed

Location	Species	Age	Seropositivity	Total
Mossaka (N = 144)	Mops condylurus	Adult	6.1% (2/33)	5.7% (2/35)
		Juvenile	0 (0/2)	
	Chaerephon pumilus	Adult	9.1% (2/22)	3.3% (2/61)
		Juvenile	0 (0/39)	
	Molossinae	Adult	13.9% (5/36)	12.5% (6/48)
		Juvenile	8.3% (1/12)	
Liranga (N = 28)	Epomops franqueti	Adult	15.4% (2/12)	15.4% (2/12)
		Juvenile	-	
	Micropteropus pusillus	Adult	0 (0/8)	0 (0/12)
		Juvenile	0 (0/4)	
	Molossinae	Adult	-	0 (0/1)
		Juvenile	0 (0/1)	
	Hypsignathus monstrosus	Adult	0 (0/1)	66.7% (2/3)
		Juvenile	100% (2/2)	
	Total	Adult	9.8% (11/112)	8.1% (14/172)
		Juvenile	5% (3/60)	

Table 1 Canalagiaal status (IgC) against EDOV	alwaannatain in bata aannalad in DOC naan tha Bilyana	DDC Ebala winned diseases such most in 2010
Table 1. Serological status (19(4) against EDUV	glycoprotein in bats sampled in ROC near the Bikoro,	DRU, EDOIA VIFUS DISEASE OULDFEAK IN 2018.

https://doi.org/10.1371/journal.pntd.0010504.t001

throughout the region and are commonly associated with human dwellings [24]. We observed the bats in the wall cavities of buildings including homes and the local hospital, suggesting opportunity for indirect contact with these bats at the human-animal interface.

Serological assays to detect Ebolavirus exposure history in bats are complicated by crossreactivity [25,26,27], a lack of virus neutralizing antibodies generated in bats against filoviruses [28,29], and a short-lived detectable antibody response in bats following viral infection [29]. Gryseels and Mbala-Kingebeni *et al.* (2020) conducted extensive surveillance following the Likati EVD outbreak in DRC and reported no seropositive samples among rodents or bats using a multiplexed Luminex assay to broadly detect exposure to filovirus antigens [30]. Though our anti-EBOV IgG ELISA is likely to be less conservative and prone to cross-reactivity with exposure to related viral antigens (e.g. Bombali virus glycoprotein) than the multiplexed Luminex assays, our results are consistent with those of the Lacroix *et al.* (2021) study which likewise reported detection of seropositive bats sampled contemporaneously near the Bikoro 2018 EVD outbreak [23].

Responding rapidly to EVD outbreak situations in remote regions is a challenging endeavor and it is often months following the index case of an outbreak before an investigation of the initial spillover can take place. Despite rapid deployment of two independent field teams to conduct surveillance near the location of an EBOV spillover event, neither team recovered evidence of active infection in bats though both teams were able to detect antibodies against Ebolaviruses in bats in the region. There remain serious obstacles to uncovering the reservoir of Ebolaviruses in central Africa including 1) sampling deeply among relevant taxa in one of the most biodiverse regions on earth and 2) sampling during periods of viral shedding in wildlife populations which may be short-lived or associated with bat birthing pulses [3,31]. With an incubation period of 2–21 days after exposure and before symptom onset in humans [32], reactive sampling following declaration of an EVD outbreak may be asynchronous with a pulse of viral shedding in the wildlife reservoir/s.

In recent years, ecological data have been successfully integrated into trait-based statistical models to predict reservoir hosts for pathogens and optimize sampling strategies [33,34,35],

though seasonality and timing of shedding events in wildlife remain less studied. Longitudinal surveys in wildlife populations including ecological context, seasonal variation, and improved understanding of regional human-domestic animal-wildlife interfaces are needed to refine sampling strategies to identify wildlife reservoirs of Ebolaviruses in the future. We emphasize the value of investment into in-country infrastructure and training for conducting surveillance activities and promoting early detection of spillover events as a critical component in both identifying periods and activities where spillover risk is highest and in mitigating outbreaks early following a spillover event [36,37].

Supporting information

S1 Table. Bat swab, blood, and tissue samples tested by qRT-PCR and Pan-filovirus RT-PCR.

(TIFF)

S2 Table. ELISA data for the seropositive samples and plate-specific serum controls collected from naïve Rousettus bats. (TIFF)

S1 Fig. An opening (approximately 15 cm in length and 15 cm in height with wires covering part of the opening) in the wall of a hospital electrical room in Mossaka, Republic of the Congo, where insectivorous bats were found roosting. (TIFF)

S1 Video. Bats exiting a roost in the walls of a structure in Mossaka, Republic of the Congo.

(MOV)

Acknowledgments

We thank the embassy of the United States in Brazzaville, World Health Organization-Republic of Congo, the staff of the Mossaka hospital and local Red Cross volunteers, and Emilie Fairet for logistical support during the field mission.

Disclaimer

Any opinions and conclusions are those of the author(s) and do not represent the official views of the U.S. Fish and Wildlife Service or the National Institutes of Health.

Author Contributions

Conceptualization: Stephanie N. Seifert, Jean-Vivien Mombouli, Fabien Roch Niama, Vincent J. Munster.

Data curation: Stephanie N. Seifert, Robert J. Fischer.

Formal analysis: Stephanie N. Seifert, Robert J. Fischer.

- **Funding acquisition:** Francine Ntoumi, César Muñoz-Fontela, Jean-Vivien Mombouli, Chris Walzer, Fabien Roch Niama, Vincent J. Munster.
- **Investigation:** Stephanie N. Seifert, Robert J. Fischer, Eeva Kuisma, Cynthia Badzi Nkoua, Gerard Bounga, Marc-Joël Akongo, Jonathan E. Schulz, Beatriz Escudero-Pérez, Beal-Junior Akoundzie, Vishnou Reize Bani Ampiri, Ankara Dieudonne, Ghislain Dzeret

Indolo, Serge D. Kaba, Igor Louzolo, Lucette Nathalie Macosso, Yanne Mavoungou, Valchy Bel-bebi Miegakanda, Rock Aimé Nina, Kevin Tolovou Samabide, Vincent J. Munster.

- Methodology: Stephanie N. Seifert, Eeva Kuisma, Gerard Bounga, Marc-Joël Akongo, Jonathan E. Schulz, Beatriz Escudero-Pérez.
- **Project administration:** Stephanie N. Seifert, Eeva Kuisma, Cynthia Badzi Nkoua, Gerard Bounga, Serge D. Kaba, Alain I. Ondzie, Sarah H. Olson, Fabien Roch Niama.
- Resources: Vincent J. Munster.
- Supervision: Stephanie N. Seifert, Robert J. Fischer, Cynthia Badzi Nkoua, Gerard Bounga, Marc-Joël Akongo, Beatriz Escudero-Pérez, Alain I. Ondzie, Francine Ntoumi, César Muñoz-Fontela, Jean-Vivien Mombouli, Sarah H. Olson, Chris Walzer, Fabien Roch Niama, Vincent J. Munster.
- Validation: Robert J. Fischer, Jonathan E. Schulz.
- Writing original draft: Stephanie N. Seifert.
- Writing review & editing: Stephanie N. Seifert, Robert J. Fischer, Eeva Kuisma, Cynthia Badzi Nkoua, Gerard Bounga, Marc-Joël Akongo, Jonathan E. Schulz, Beatriz Escudero-Pérez, Beal-Junior Akoundzie, Vishnou Reize Bani Ampiri, Ghislain Dzeret Indolo, Serge D. Kaba, Igor Louzolo, Lucette Nathalie Macosso, Yanne Mavoungou, Valchy Bel-bebi Miegakanda, Kevin Tolovou Samabide, Alain I. Ondzie, Francine Ntoumi, César Muñoz-Fontela, Jean-Vivien Mombouli, Sarah H. Olson, Chris Walzer, Fabien Roch Niama, Vincent J. Munster.

References

- 1. Leroy EM, Kumulungui B, Pourrut X, Rouquet P, Hassanin A, Yaba P, et al. Fruit bats as reservoirs of Ebola virus. Nature. 2005; 438(7068):575–6. https://doi.org/10.1038/438575a PMID: 16319873
- Amman BR, Bird BH, Bakarr IA, Bangura J, Schuh AJ, Johnny J, et al. Isolation of Angola-like Marburg virus from Egyptian rousette bats from West Africa. Nat Commun. 2020; 11(1):510. <u>https://doi.org/10. 1038/s41467-020-14327-8</u> PMID: 31980636
- Amman BR, Carroll SA, Reed ZD, Sealy TK, Balinandi S, Swanepoel R, et al. Seasonal pulses of Marburg virus circulation in juvenile Rousettus aegyptiacus bats coincide with periods of increased risk of human infection. PLoS Pathog. 2012; 8(10):e1002877. https://doi.org/10.1371/journal.ppat.1002877 PMID: 23055920
- Goldstein T, Anthony SJ, Gbakima A, Bird BH, Bangura J, Tremeau-Bravard A, et al. The discovery of Bombali virus adds further support for bats as hosts of ebolaviruses. Nat Microbiol. 2018; 3(10):1084–9. https://doi.org/10.1038/s41564-018-0227-2 PMID: 30150734
- Jones MEB, Amman BR, Sealy TK, Uebelhoer LS, Schuh AJ, Flietstra T, et al. Clinical, Histopathologic, and Immunohistochemical Characterization of Experimental Marburg Virus Infection in A Natural Reservoir Host, the Egyptian Rousette Bat (Rousettus aegyptiacus). Viruses. 2019; 11(3).
- Negredo A, Palacios G, Vazquez-Moron S, Gonzalez F, Dopazo H, Molero F, et al. Discovery of an ebolavirus-like filovirus in europe. PLoS Pathog. 2011; 7(10):e1002304. <u>https://doi.org/10.1371/journal.ppat.1002304</u> PMID: 22039362
- Yang XL, Tan CW, Anderson DE, Jiang RD, Li B, Zhang W, et al. Characterization of a filovirus (Mengla virus) from Rousettus bats in China. Nat Microbiol. 2019; 4(3):390–5. <u>https://doi.org/10.1038/s41564-018-0328-y PMID: 30617348</u>
- Nanclares C, Kapetshi J, Lionetto F, de la Rosa O, Tamfun JJ, Alia M, et al. Ebola Virus Disease, Democratic Republic of the Congo, 2014. Emerg Infect Dis. 2016; 22(9):1579–86. https://doi.org/10.3201/ eid2209.160354 PMID: 27533284
- Ebola Outbreak Epidemiology T. Outbreak of Ebola virus disease in the Democratic Republic of the Congo, April-May, 2018: an epidemiological study. Lancet. 2018; 392(10143):213–21. https://doi.org/ 10.1016/S0140-6736(18)31387-4 PMID: 30047375
- 10. Mbala-Kingebeni P, Villabona-Arenas CJ, Vidal N, Likofata J, Nsio-Mbeta J, Makiala-Mandanda S, et al. Rapid Confirmation of the Zaire Ebola Virus in the Outbreak of the Equateur Province in the

Democratic Republic of Congo: Implications for Public Health Interventions. Clin Infect Dis. 2019; 68 (2):330–3. https://doi.org/10.1093/cid/ciy527 PMID: 29961823

- Nachega JB, Mbala-Kingebeni P, Otshudiema J, Mobula LM, Preiser W, Kallay O, et al. Responding to the Challenge of the Dual COVID-19 and Ebola Epidemics in the Democratic Republic of Congo-Priorities for Achieving Control. Am J Trop Med Hyg. 2020; 103(2):597–602. <u>https://doi.org/10.4269/ajtmh.</u> 20-0642 PMID: 32563272
- Mari Saez A, Weiss S, Nowak K, Lapeyre V, Zimmermann F, Dux A, et al. Investigating the zoonotic origin of the West African Ebola epidemic. EMBO Mol Med. 2015; 7(1):17–23. <u>https://doi.org/10.15252/emmm.201404792 PMID</u>: 25550396
- Forbes KM, Webala PW, Jaaskelainen AJ, Abdurahman S, Ogola J, Masika MM, et al. Bombali Virus in Mops condylurus Bat, Kenya. Emerg Infect Dis. 2019; 25(5):955–7. https://doi.org/10.3201/eid2505. 181666 PMID: 31002301
- Karan LS, Makenov MT, Korneev MG, Sacko N, Boumbaly S, Yakovlev SA, et al. Bombali Virus in Mops condylurus Bats, Guinea. Emerg Infect Dis. 2019; 25(9). https://doi.org/10.3201/eid2509.190581 PMID: 31310231
- Haddock E, Feldmann F, Feldmann H. Effective chemical inactivation of Ebola virus. Emerg Infect Dis. 2016; 22(7):1292–1294. https://doi.org/10.3201/eid2207.160233 PMID: 27070504
- de Wit E, Rosenke K, Fischer RJ, Marzi A, Prescott J, Bushmaker T, et al. Ebola Laboratory Response at the Eternal Love Winning Africa Campus, Monrovia, Liberia, 2014–2015. J Infect Dis. 2016; 214 (suppl 3):S169–S76. https://doi.org/10.1093/infdis/jiw216 PMID: 27333914
- Schulz JE, Seifert SN, Thompson JT, Avanzato V, Sterling SL, Yan L, et al. Serological Evidence for Henipa-like and Filo-like Viruses in Trinidad Bats. J Infect Dis. 2020; 221(Suppl 4):S375–S82. https:// doi.org/10.1093/infdis/jiz648 PMID: 32034942
- Ducroz JF, Volobouev V, Granjon L. An Assessment of the Systematics of Arvicanthine Rodents Using Mitochondrial DNA Sequences: Evolutionary and Biogeographical Implications. Journal of Mammalian Evolution. 2001; 8(3):173–206.
- Changula K, Kajihara M, Mori-Kajihara A, Eto Y, Miyamoto H, Yoshida R, et al. Seroprevalence of Filovirus Infection of Rousettus aegyptiacus Bats in Zambia. J Infect Dis. 2018; 218(suppl_5):S312–S7. https://doi.org/10.1093/infdis/iiy266 PMID: 29889270
- Leendertz SA, Gogarten JF, Dux A, Calvignac-Spencer S, Leendertz FH. Assessing the Evidence Supporting Fruit Bats as the Primary Reservoirs for Ebola Viruses. Ecohealth. 2016; 13(1):18–25. <u>https://doi.org/10.1007/s10393-015-1053-0 PMID: 26268210</u>
- Pourrut X, Souris M, Towner JS, Rollin PE, Nichol ST, Gonzalez JP, et al. Large serological survey showing cocirculation of Ebola and Marburg viruses in Gabonese bat populations, and a high seroprevalence of both viruses in Rousettus aegyptiacus. BMC Infect Dis. 2009; 9:159. <u>https://doi.org/10.1186/</u> 1471-2334-9-159 PMID: 19785757
- 22. De Nys HM, Kingebeni PM, Keita AK, Butel C, Thaurignac G, Villabona-Arenas CJ, et al. Survey of Ebola Viruses in Frugivorous and Insectivorous Bats in Guinea, Cameroon, and the Democratic Republic of the Congo, 2015–2017. Emerg Infect Dis. 2018; 24(12):2228–40. https://doi.org/10.3201/eid2412. 180740 PMID: 30307845
- 23. Lacroix A, Mbala Kingebeni P, Ndimbo Kumugo SP, Lempu G, Butel C, Serrano L, et al. Investigating the Circulation of Ebola Viruses in Bats during the Ebola Virus Disease Outbreaks in the Equateur and North Kivu Provinces of the Democratic Republic of Congo from 2018. Pathogens. 2021; 10(5):557. https://doi.org/10.3390/pathogens10050557 PMID: 34064424
- Bouchard S. Sex discrimination and roostmate recognition by olfactory cues in the African bats, Mops condylurus and Chaerephon pumilus (Chiroptera: Molossidae). Journal of Zoology. 2001; 254 (1):109–17.
- Nakayama E, Yokoyama A, Miyamoto H, Igarashi M, Kishida N, Matsuno K, et al. Enzyme-linked immunosorbent assay for detection of filovirus species-specific antibodies. Clin Vaccine Immunol. 2010; 17 (11):1723–8. https://doi.org/10.1128/CVI.00170-10 PMID: 20861331
- Schuh AJ, Amman BR, Sealy TS, Flietstra TD, Guito JC, Nichol ST, et al. Comparative analysis of serologic cross-reactivity using convalescent sera from filovirus-experimentally infected fruit bats. Sci Rep. 2019; 9(1):6707. https://doi.org/10.1038/s41598-019-43156-z PMID: 31040343
- Ogawa H, Miyamoto H, Nakayama E, Yoshida R, Nakamura I, Sawa H, et al. Seroepidemiological Prevalence of Multiple Species of Filoviruses in Fruit Bats (Eidolon helvum) Migrating in Africa. J Infect Dis. 2015; 212 Suppl 2:S101–8. https://doi.org/10.1093/infdis/jiv063 PMID: 25786916
- Schuh AJ, Amman BR, Sealy TK, Kainulainen MH, Chakrabarti AK, Guerrero LW, et al. Antibody-Mediated Virus Neutralization Is Not a Universal Mechanism of Marburg, Ebola, or Sosuga Virus Clearance in Egyptian Rousette Bats. J Infect Dis. 2019; 219(11):1716–21. https://doi.org/10.1093/infdis/jiy733 PMID: 30590775

- Schuh AJ, Amman BR, Sealy TK, Spengler JR, Nichol ST, Towner JS. Egyptian rousette bats maintain long-term protective immunity against Marburg virus infection despite diminished antibody levels. Sci Rep. 2017; 7(1):8763. https://doi.org/10.1038/s41598-017-07824-2 PMID: 28821722
- Gryseels S, Mbala-Kingebeni P, Akonda I, Angoyo R, Ayouba A, Baelo P, et al. Role of Wildlife in Emergence of Ebola Virus in Kaigbono (Likati), Democratic Republic of the Congo, 2017. Emerg Infect Dis. 2020; 26(9):2205–9. https://doi.org/10.3201/eid2609.191552 PMID: 32818404
- Reed Hranac C, Marshall JC, Monadjem A, Hayman DTS. Predicting Ebola virus disease risk and the role of African bat birthing. Epidemics. 2019; 29:100366. https://doi.org/10.1016/j.epidem.2019.100366 PMID: 31744768
- Geisbert TW, Jahrling PB. Exotic emerging viral diseases: progress and challenges. Nat Med. 2004; 10 (12 Suppl):S110–21. https://doi.org/10.1038/nm1142 PMID: 15577929
- Becker DJ, Seifert SN, Carlson CJ. Beyond Infection: Integrating Competence into Reservoir Host Prediction. Trends Ecol Evol. 2020; 35(12):1062–5. <u>https://doi.org/10.1016/j.tree.2020.08.014</u> PMID: 32921517
- Han BA, Schmidt JP, Alexander LW, Bowden SE, Hayman DT, Drake JM. Undiscovered Bat Hosts of Filoviruses. PLoS Negl Trop Dis. 2016; 10(7):e0004815. <u>https://doi.org/10.1371/journal.pntd.0004815</u> PMID: 27414412
- Becker DJ, Albery GF, Sjodin AR, Poisot T, Dallas TA, Eskew EA, et al. Predicting wildlife hosts of betacoronaviruses for SARS-CoV-2 sampling prioritization: a modeling study. bioRxiv. 2020:2020.05.22.111344.
- Munster VJ, Bausch DG, de Wit E, Fischer R, Kobinger G, Munoz-Fontela C, et al. Outbreaks in a Rapidly Changing Central Africa—Lessons from Ebola. N Engl J Med. 2018; 379(13):1198–201. https://doi. org/10.1056/NEJMp1807691 PMID: 30134126
- 37. Kuisma E, Olson SH, Cameron KN, Reed PE, Karesh WB, Ondzie AI, Akongo MJ, Kaba SD, Fischer RJ, Seifert SN, Muñoz-Fontela C, Becker-Ziaja B, Escudero-Pérez B, Badzi Nkoua C, Munster VJ, and Mombouli JV. Long-term wildlife mortality surveillance in northern Congo: a model for the detection of Ebola virus disease epizootics. Phil Trans R Soc B. 2019; 374: 20180339 https://doi.org/10.1098/rstb. 2018.0339 PMID: 31401969